\documentclass[aspectratio=169]{beamer}
\title{Presentation 2}
\subtitle{Progress Update}
\author{CS460/660 - Machine Learning}
\date{14--15 Sep 2026}
\begin{document}
\begin{frame}{Base paper and problem setup}
\begin{itemize}
  \item Base paper and motivation
  \item What is already solved?
  \item Input and output
  \item Dataset and preprocessing
\end{itemize}
\end{frame}
\begin{frame}{Research question and tentative timeline}
\begin{itemize}
  \item What problem are you asking?
  \item Scope, assumptions, and expected contribution
  \item Tentative deliverables and dates
\end{itemize}
\end{frame}
\begin{frame}{What did you try?}
\begin{itemize}
  \item Method, baseline, or experiment tried
  \item Result or observation
  \item What changed after the attempt?
\end{itemize}
\end{frame}
\begin{frame}{Why did you try it?}
\begin{itemize}
  \item Motivation for the choice
  \item Expected outcome
  \item What worked, failed, or remains uncertain?
\end{itemize}
\end{frame}
\begin{frame}{Updated timeline}
\begin{itemize}
  \item Completed deliverables
  \item Revised milestones and dates
  \item Next experiment and plan for Presentation 3
\end{itemize}
\end{frame}
\end{document}
